Author’s response to reviews

Title: Correlation between the Charlson comorbidity index and skeletal muscle mass/physical performance in hospitalized older people potentially suffering from sarcopenia

Authors:

Ge Gong (gonggedoc@163.com)
Wenhui Wan (356133926@qq.com)
Xinghu Zhang (zhangxh197002@163.com)
Yu Liu (liuyufanwei@163.com)
Xinhui Liu (linxinhuijnyy@163.com)
Jian Yin (yinjiandoc@163.com)

Version: 5 Date: 14 Dec 2019

Author’s response to reviews:

Thank you for your letter and for the comments.

1. Competing interests (This section refers to any any financial or non-financial competing interests. If there are no competing interests, please state “The authors declare no competing interests”.)
   Response to the comment: We have modified the competing interests and the amendment can be found in Declarations,line 14,page 13.

2. Consent to participate (As your study is of retrospective design, consent to participate is not required for your study. Instead, please state who provided administrative consent to access and use the medical records described in your study.)
   Response to the comment: We have deleted the consent to participate and have stated the Ethical Committee provided consent to access the medical records in Declarations,line 8,page 13.